Author's response to reviews

Title: Endoscopist specialty is associated with colonoscopy quality

Authors:

Mengzhu Jiang (mengzhu.jiang@mcgill.ca)
Maida J Sewitch (maida.sewitch@mcgill.ca)
Alan N Barkun (alan.barkun@muhc.mcgill.ca)
Lawrence Joseph (lawrence.joseph@mcgill.ca)
Robert J Hilsden (rhilsden@ucalgary.ca)

Version: 2 Date: 28 August 2012

Author's response to reviews:

In response to your suggestion, we have added the sentence, 'Ethics approval was obtained from McGill University Faculty of Medicine Institutional Review Board and research ethics boards at each study site.' to the revised manuscript (last sentence in Study Design section (page 4)).